Author’s response to reviews

Title: Impact of drug reconciliation at discharge and communication between hospital and community pharmacists on drug-related problems: study protocol for a cluster randomized cross-over trial

Authors:

Xavier Pourrat (x.pourrat@chu-tours.fr)
Clarisse Roux (clarisse.roux@chu-nimes.fr)
Brigitte Bouzige (Bouzige.brigitte@orange.fr)
Valérie garnier (pharmacie-garnier@wanadoo.fr)
Armelle Develay (armelle.develay@chu-nimes.fr)
Benoit Allenet (BAllenet@chu-grenoble.fr)
Martial Fraysse (martial.fraysse@wanadoo.fr)
Jean-Michel Halimi (jean-michel.halimi@univ-tours.fr)
Jacqueline Grassin (j.grassin@chu-tours.fr)
Bruno Giraudreau (bruno.giraudreau@univ-tours.fr)

Version: 2  Date: 18 April 2014

Author’s response to reviews:

I added the name of the ethics committee that give the authorization for the study for all centers.
also I modified the status of the study at today.
I send by mail to the editor the administrative documentation and his english translation